Author's response to reviews

Title: Preliminary study of p53 and c-erbB-2 expression in gallbladder cancer in Indian patients

Authors:

Amita Chaube (amitachaube@yahoo.com.in)
Mallika Tewari (mallika_vns@satyam.net.in)
R S Garbyal (raigarbyal@yahoo.co.in)
Usha Singh (harishukla@usa.net)
Hari S Shukla (harishukla@usa.net)

Version: 10  Date: 6 May 2006

Author’s response to reviews: see over
Title: [manuscript id. 8962091628764582]

Preliminary study of p53 and c-erbB-2 expression in gallbladder cancer in Indian patients

Authors:

Amita Chaube (amitachaube@yahoo.com.in)
Mallika Tewari (mallika_vns@satyam.net.in)
Rajendra S Garbyal (rajgarbyal@yahoo.co.in)
Usha Singh (harishukla@usa.net)
Hari S Shukla (harishukla@usa.net)

Date: 6th May 2006

Subject: Formatting Changes for MS: [8962091628764582]

Dear Sir,

Thank you for your email. Below are the point-by-by answers to the questions raised:

Formatting changes requested were:
-----------------------------------------------
Title & Abstract - Please ensure these are identical in the manuscript and in the submission system.

Title & abstract have been checked. Ok.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Checked. OK.

Please take note of editorial corrections made to the manuscript.

All editorial corrections have been incorporated in the manuscript.
Please check authors' names in the submission system. Is it Mallika Tewari (submissions system) or Tewari Mallika (Manuscript)?

It is Mallika Tewari. Now stands so in both the manuscript and submission system.

Patients & Methods - Please rename this section 'Methods'.

Renamed as suggested

Competing interests - If there are none to declare, please write only 'The authors declare that they have no competing interests'.

Done as suggested

I have made the necessary corrections in the manuscript as suggested and I look forward to its early publication.

Thanking you for your kind co-operation.

Kind regards,

Hari S Shukla
Professor & Head
Department of Surgical Oncology
Institute of Medical Sciences
Banaras Hindu University
Varanasi – 221005
U.P. (INDIA)
Tel: 0091 542 236 7718
Fax: 0091 542 236 8856
Email: harishukla@usa.net